Author’s response to reviews

Title: A cerebral Arteriovenous Malformation mistakenly diagnosed as Dry Eye and Glaucoma: a case report

Authors:

Shuhao Shen (jnussh@hotmail.com)
Xiaoyong Liu (jndxlxy@163.com)
Jian Chen (drchenj@163.com)
Chengyou Yang (835013859@qq.com)
Changzheng Shi (tsczcn@jnu.edu.cn)
Qing Zhou (kerryzh@163.com)

Version: 1 Date: 05 Jun 2019

Author’s response to reviews:

First of all, thanks for the reviewer's consultation and Suggestions. I will modify the paper as follows according to your Suggestions:

1. We improved the photos of the anterior segment of both eyes and tried to clearly explain the special features of ocular congestion caused by AVM, so as to facilitate the diagnosis by clinicians.

2. Fundus photography was added to rule out lesions in the optic disc. However, we were not able to provide relevant examination results to support the diagnosis since the patient did not have visual field examination.

3. Corrected inappropriate words and grammatical errors in the text.

4. As suggested by your editor, the full text will be submitted to American Journal Experts for English polishing to meet the reading habits and academic requirements of the readers.

The above is our revision of the paper according to the comments of reviewers.

We hope that such rare cases can be more widely known by readers of BMC ophthalmology.

If this change cannot fully meet the requirements of BMC, I hope to receive your Suggestions and I will try my best to improve it.